Author’s response to reviews

Title: How do stakeholders from multiple hierarchical levels of a large provincial health system define engagement? A qualitative study

Authors:
Jill Norris (jmorris@ucalgary.ca)
Deborah White (dwhit@ucalgary.ca)
Lorelli Nowell (lnowell@ucalgary.ca)
Kelly Mrklas (kelly.mrklas@albertahealthservices.ca)
Henry Stelfox (tstelfox@ucalgary.ca)

Version: 4 Date: 20 Jun 2017

Author’s response to reviews:
Handling editor's comment:
Please note that you have repeated the term issue twice in the revised sentence in the abstract and change this. The term issue is I find confusing the term problem or matter is perhaps an alternative.

We revised the sentence to read, "Second, engagement centered on a shared decision-making process about meaningful change for everyone “around the table,” those who are most impacted."